C

O

S

4

2 3 , F e b r u ar y 22,1

99 6

Due

:

Mar

c

h 7, 1996 Ha n

d ou t

Nu

mber5

H o

m e

wor

k Se

t

No

.

2

R e a d i n g A

ssign

men

ts

Read

[C

LR

] Chapter 25.2. Read [K] Chapters 5,8

and 9. Optiona l Readings: [K] Chapters 10 and 11; [CLR] Chapters 20-22

as needed. W ritten Assignments A. DoProblem 2-4.B.

Do Problem 20-2

C. Do Problem 21-1.

D. Do Problem 22-1.E.

Do Problem 24-1.

F. Do Problem 25-2.(

Y ou maywant to consult the material in Chapter

25.4)

G. Do Problem 25-4.

H. Do Exercise 21.4-2.I.

Do Exercise 25.2-2.J.

Do Exercise 25.2-4.

Remar ks Y ou should be able to start working on A, D, E after today’s

class, and the rest after nextT uesday’s class.1